Author's response to reviews

Title: Role of malignant ascites on human mesothelial cells and their gene expression profiles

Authors:

Isabelle Matte (isabelle.matte@usherbrooke.ca)
Denis Lane (denis.lane@usherbrooke.ca)
Dimcho Bachvarov (dimtchho.bachvarov@crhdq.ulaval.ca)
Claudine Rancourt (claudine.rancourt@usherbrooke.ca)
Alain Piche (alain.piche@usherbrooke.ca)

Version: 8 Date: 27 March 2014

Author's response to reviews: see over
BMC Cancer
Editorial office

Please find enclosed a copy of our revised manuscript entitled “Role of malignant ascites on human mesothelial cells and their gene expression profiles” for consideration in BMC Cancer.

The editorial request has been addressed. The ethic statement on page 23 has been removed. On line 109, we have replaced the phrase by the following statement: This study has been performed in accordance with the Declaration of Helsinki and was approved by the «Comité d’éthique de la recherche en santé chez l’humain du centre hospitalier universitaire de Sherbrooke».

We appreciate the careful review of our manuscript provided by the Editors and the reviewers. We hope that our revised manuscript will be processed without any further delay.

Yours sincerely,

Alain Piché M.D, M.Sc.
Professor
Dept Microbiologie et Infectiologie
Université de Sherbrooke
Québec, Canada